

Available online at www.CivileJournal.org

# **Civil Engineering Journal**

Vol. 6, Special Issue "Emerging Materials in Civil Engineering", 2020



**Review Article** 

## A Review of the Utilisation of Recycled Waste Material as an Alternative Modifier in Asphalt Mixtures

## Abdalrhman Milad <sup>a\*</sup>, Ahmed Suliman B. Ali <sup>b</sup>, Nur Izzi Md Yusoff <sup>a</sup>

<sup>a</sup> Department of Civil Engineering, Faculty of Engineering and Built Environment, Universiti Kebangsaan Malaysia, Selangor, Malaysia.

<sup>b</sup> Department of Civil & Structural Engineering, UTHM, 86400 Parit Raja, Batu Pahat, Malaysia.

Received 22 July 2020; Accepted 07 October 2020

#### Abstract

The possibility of using waste materials in road construction is of great interest as their utilisation may contribute to reducing the problems of hazard and pollution and conserve natural resources. Thus, there is an urgent need to find a sustainable method for using waste materials as a substitute in the standard asphalt binders. There are several concerns about the physical and chemical properties and mechanical performance of asphalt pavements incorporated with waste material in the effort to reduce permanent deformation of the road surface. This review article presents a brief discussion of the asphalt mixtures modified with waste material, and the recycled materials used as a modifier in the asphalt mixture. The present paper summarises the use of crumb rubber, crushed concrete, steel slag, glass fibre and plastic waste in asphalt mixtures. The use of waste materials as a modifier in asphalt mixture resulted in improved asphalt pavement performance. Results advocate that rubberised asphalt mixture with desired properties can be designed as an additive with a friendly environmental approach in construction materials. The researches that adopted the influence of usage, recycle waste material to improve the performance of the asphalt of the road are still limited compared to other construction fields.

Keywords: Waste Materials; Properties; Asphalt Modifier; Asphalt Pavement; Recycling.

## 1. Introduction

In developed countries, the waste materials generated by the industries pose a serious environmental problem by manufacturers or individuals that lack any residual value is termed as waste [1, 2]. The feasibility engorging waste material used in the highways construction industry is playing a leading role worldwide as a green investment movement [3, 4]. In recent years researchers have been carried out to explore the possibility of using alternative waste material such as steel slag, plastic and scrap tires to construct Hot-Mix Asphalt (HMA) pavements to minimise the deterioration of road pavements [3]. In 2018, the total amount of waste generated, including those generated by households and economic activities in 28 European Union countries, exceed 2.5 billion tons. A total of 285 million tons of scrap tires are discarded in the USA yearly, as an example, as well as used in Portugal, China, Canada, and South Africa [5, 6]. Recycled scrap tires can be used as raw materials in the form or crumb rubber and powder rubber [7]. Research has shown that waste material such as rubber tires is suitable for incorporation in asphalt mixture since rubber tire can improve the performance of asphalt pavement and reduce the need to use new raw materials for the construction of road pavements [8-10]. Furthermore, the crushed concrete in several types of research confirmed that the recycled concrete aggregates (RCA) are applicable for reuse on the roads, concrete pavement when it is replaced

\* Corresponding author: miladabdalrhman@siswa.ukm.edu.my

doi) http://dx.doi.org/10.28991/cej-2020-SP(EMCE)-05



© 2020 by the authors. Licensee C.E.J, Tehran, Iran. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC-BY) license (http://creativecommons.org/licenses/by/4.0/).

with a proper percentage of new aggregate [11-14]. The use of recycled materials as an eco-friendly substitute has performance, economic and environmental benefits [15, 16]. Figure 1 shows the waste hierarchy applied to pavement materials. The most preferred choice is reducing the production of waste materials, whereas landfilling is the least preferable choice [15].



Figure 1. The waste hierarchy for pavement materials [15]

Recent studies have proven the economic advantage of adding tires to asphalt binders since recycled tires markedly reduce the need for using new materials [16]. The use of waste materials for asphalt stabilisation is regarded as a smart strategy for sustainable development. Many researchers have explored the methods for improving the use of waste materials such as crumb rubber, crushed concrete, steel slag, glass fibre and plastic waste in bitumen modifier for hot asphalt mixture properties and performance of factors affecting pavement damage. Table 1 summaries some of the previous research which used industrial wastes material in modified bitumen. The present study explores the possibility of using waste material to reduce/prevent permanent deformation of road surface and enhance the durability and strength of the road pavements. Section 2 describes the method used for sources of kinds of literature and strategies. Section 3 describes the primary property requirements for waste materials. Section 4 describes waste materials as modifiers for each type. Section 5 describes the challenges in incorporating waste materials in asphalt pavements, and Section 6 discusses the environmental and economic aspects. Finally, Section 7 presents the conclusions and final remarks.

| Table 1. | Summary | review on | the wastes | s materials us | ed to modi | fiers in as | ohalt mixture |
|----------|---------|-----------|------------|----------------|------------|-------------|---------------|
|          |         |           |            |                |            |             |               |

| Author & Year                  | Additives by Percentage   | Material Used                              | Test  | Findings and Summary  |
|--------------------------------|---------------------------|--|---|---|
| Mashaan et al.<br>2013[17]     | 6, 12, 16, and 20%        | Crumb rubber                               | <ol> <li>Marshall test</li> <li>Wheel tracking</li> <li>Indirect tensile</li> </ol>   | <ol> <li>Better stability</li> <li>Better adhesion</li> <li>The best enhancement was 12% in the wet process.</li> </ol>   |
| Arabani et al.,<br>2018 [18]   | 1, 3 and 5%               | Crumb rubber                               | 1-Marshal tests   | <ol> <li>Improved stability.</li> <li>Ideal percentage is 1%.</li> </ol>  |
| Pasandín et al.,<br>2017 [19]  | 0, 35 and 42%             | Crushed concrete<br>& Waste tire<br>rubber | 1-Indirect tensile fatigue<br>test (ITFT)   | 1-Improvement in fatigue performance<br>2-HMA containing RCA are suitable for medium<br>and light traffic because of its low resistance to<br>the fragmentation of RCA. |
| Ahmadinia et al.,<br>2012 [20] | 0, 2, 4, 6, 8 and 10%     | Waste plastic<br>bottles                   | <ol> <li>Resilient modulus test</li> <li>Drain down test</li> <li>Wheel tracking test</li> <li>Tensile strength test</li> </ol> | <ul><li>1-Optimum PET is 4 and 6% by weight of optimum bitumen content</li><li>2- PET has better rutting resistance than the ordinary mixture.</li></ul>                |
| Ziari et al., 2016<br>[21]     | 0, 0.25, 0.5, 0.75 and 1% | Waste plastic bottle (PET)                 | <ol> <li>Dynamic creep test</li> <li>Wheel tracking test</li> </ol>   | 1-Reduced deformation on rutting.   |

| Author & Year                     | Additives by Percentage | Material Used                                | Test   | Findings and Summary  |
|-----------------------------------|-------------------------|--|--|---|
| White 2019 [22]                   | 6%                      | 1- Plastic bags<br>2-Waste plastic<br>bottle | <ol> <li>Indirect tensile</li> <li>fatigue test (ITFT)</li> <li>Wheel tracking test</li> </ol>                                   | 1- Enhanced deformation resistance  |
| Fu et al., 2017<br>[23]           | 0 to 0.8%               | Glass fibre                                  | 1-Viscosity test<br>2- Direct tensile test   | <ol> <li>Improved crack resistance.</li> <li>Improvement effects positively permeability.</li> <li>High cost compare to the use of virgin asphalt.</li> </ol>   |
| Celauro &<br>Pratico 2018<br>[24] | 0 and 0.3%              | Basalt fibre                                 | 2-Marshall test  | <ol> <li>Better rutting resistance.</li> <li>Environmentally-friendly.</li> <li>Cost effective</li> </ol>   |
| Wu. et al., 2014<br>[25]          | 0.3% and 0.4%           | Lignin fibre<br>Carbon fibre                 | <ol> <li>Penetration test</li> <li>Viscosity test</li> <li>Cone penetration test</li> <li>Rheological<br/>measurement</li> </ol> | <ol> <li>Significant improvement in mortar's softening<br/>point, penetration, viscosity, and cone<br/>penetration.</li> <li>Optimal lignin fibre content of 0.3% and<br/>optimal carbon fibre content of 0.40%.</li> </ol>                                       |
| Maharaj et al.,<br>2017 [26]      | 0 to 20%                | Steel slag                                   | 1- Marshall stability test   | 1-The Marshall specification for the design of HMA is 15% of steel slag aggregate addition has similar features as the 0% formulation.  |
| Gowtham et al.,<br>2018 [27]      | 0 to 10%                | Steel slag                                   | 1-Marshall test  | <ol> <li>Optimum stone substitution of 10% in the mixtures.</li> <li>Increasing in the percentage of air voids and decrease in voids filled with bitumen because of the porous nature of the steel slag.</li> </ol>   |
| Li, et al., 2018<br>[28]          | 5%                      | Powder steel slag                            | <ol> <li>Extended bending<br/>beam Rheometer test</li> <li>2-Double-edge notched<br/>tension test</li> </ol>                     | <ol> <li>The addition of particles that are 5% larger can<br/>diminish the severity of reversible aging.</li> <li>In regions with cold climate, steel slag has a<br/>better low-temperature performance than<br/>conventional limestone powder system.</li> </ol> |

## 2. Sources of Literatures and Strategies

This review focuses on the performance of five types of waste materials. All waste materials are readily available in their respective study areas as defined by the researchers. Highlighting is given to identifying effects on constrictions and environmental performance of asphalt mixtures after the incorporation of waste materials. This study also deliberates the effect of physical, chemical and mechanical modification of several materials and how these modifications affect the performance of the waste materials in the binder and mixes. Finally, the present study identifies the large gaps in the current literature and recommends future research. An organised literature review was carried out to meet the research objectives stated in the previous section. Pertinent works of literature are identified using keyword-based searches in various electronic databases and library services. Figure 2 shows the sources of literature that display two primarily explored the most famous indexed databases: The web of Science database and the Google Scholar database.



Figure 2. Sources of Literature

## **3.** Property Requirements for Waste Materials

The feasibility use of waste materials comes with a list of reasonable engineering concerns such as the impact on production, strength and durability and future recyclability; environmental issues for instance processing and handling methods, leaching, emissions, fumes and economic issues for instance salvage value, life cycle costs and lack of funding [29-32]. Enhanced asphalt properties will improve the asphalt's resistance to many pavement problems. Various types of waste material are produced daily, and the volume of wastes generated continues to increase annually as consumption increases. The best method to deal with these wastes is by recycling and reusing them as modifiers or raw materials [33]. The recycled materials used in asphalt mixtures have to meet the same prerequisites as property classification. Figure 3 presents the requirements for incorporating waste materials in pavements. Lewandowski has researched and summarised the primary reasons for modifying bituminous materials with different types of additive [34]:

- To obtain softer blends at low service temperatures and reduce cracking.
- To improve the fatigue resistance of the blends.
- To increase the stability and strength of the mixtures.
- To obtain stiffer blends at high temperatures and reduce rutting.
- To reduce the structural thickness of pavements.

Finally, the modifier used in a specific project is influenced by various factors, such as construction ability, availability, cost, and expected performance. Several studies have shown that the technical reasons for incorporating modifiers in bituminous mixtures are to obtain stiffer mixtures at high service temperatures, prevent rutting, obtain softer mixtures at low service temperatures to minimise thermal cracking, and improve the fatigue resistance of asphalt pavement.



Figure 3. Requirements for incorporating Waste Materials in Pavement

## 4. Waste Materials as Modifier

## 4.1. Crumb Rubber

Disposal of used tires is a global problem as more than 50% of the tires are discarded without any treatment. About five billion more are expected to be discarded regularly by the year 2030 [35]. The use of crumb rubber as a modifier to enhance asphalt mixture is one way for ensuring sustainable development. Crumb rubber modifier (CRM) can be a substitute polymer material for enhancing the performance of HMA [5, 36]. CRM is a waste scrap tires disposed of in trash dumps, and each year about 1.5 billion tires are manufactured globally [37]. The use of crumb rubber in the design of asphalt pavements has reduced the amount of waste in the United States, where about 10% of the asphalt

pavements contain 3% scrap tire [38]. Figure 4 shows the crumb rubber waste material before and after recycling. Two methods can be used to incorporate crumb rubber into asphalt mixtures, namely wet process and dry process. Scrap tire serves as a cement modifier in asphalt while another is used as a fine aggregate [39].



Figure 4. Crumb rubber waste material before and after recycling [40]

## **4.1.1. Physical Properties**

The physical properties of modified bitumen are investigated to identify the improvement in the viscosity and softening point of bituminous binder. However, in order to attain a balanced and superior CRM low and high-temperature features, there are many factors, for example, temperature, time of mixing, source and characteristics of the bitumen and crumb rubber which need to be taken into account as these are the factors that govern the resulting performance of asphaltic mixes 80-100 penetration grade as displayed in Table 2 below.

| <b>D</b> : 1  | Percentage of rubber (by weight of binder) |     |     |     |      |      |      |  |  |  |
|---|--|-----|-----|-----|------|------|------|--|--|--|
| Binder property   | 0%   | 6%  | 9%  | 12% | 15%  | 18%  | 21%  |  |  |  |
| Viscosity at 176°C (cP) (ASTM D2669)                              | 60   | 550 | 800 | 900 | 1500 | 2500 | 6000 |  |  |  |
| Cone penetration in 0.1 mm, 150 g, 5 seconds at 25°C (ASTM D5329) | 48   | 40  | 43  | 44  | 40   | 30   | 27   |  |  |  |
| Resilience (%) at 25°C (ASTM D5329)                               | -1   | -1  | 12  | 19  | 23   | 40   | 47   |  |  |  |
| Softening point, °C (ASTM D36)                                    | 50   | 52  | 58  | 60  | 61   | 63   | 72   |  |  |  |

| Tuble 2. The physical properties of cramb rubber mounted asphare [11] | Table 2. The phys | ical properties of cr | umb rubber modifie | d asphalt [41] |
|---|-------------------|-----------------------|--------------------|----------------|
|---|-------------------|-----------------------|--------------------|----------------|

The average loose density of tire shreds varies between 390 kg/m<sup>3</sup> to 535 kg/m<sup>3</sup> depending on the size of the shreds, while the compacted density range between 650 kg/m<sup>3</sup> to 840 kg/m<sup>3</sup> [42]. The stability, density, flow, air voids, and stiffness of HMA are markedly enhanced with the incorporation of up to 18% CR material by weight of optimum content significantly improved [43, 44]. Several researchers suggested the improvement of high-temperature long term storage stability without phase separation [45]. The incorporation of crumb rubber into asphalt mixture resulted in enhanced rutting and fatigue resistance, where the optimal percentage is 12% crumb rubber by weight of bitumen [17]. The incorporation of cryogenic rubber into asphalt mixtures can improve stability and reduce flow, improving fatigue resistance, rutting resistance, and low-temperature cracking.

## 4.1.2. Chemical Properties

The chemical properties of CR have a noticeable significant effect on asphalt, and each displays the nature of crude oil utilised to formulate it [46]. Asphalt consists of small amounts of various metals and hydrocarbons. Under normal atmospheric conditions, oxygen affects the weak bonding of the asphalt molecules and transform into water and carbon dioxide due to complete oxidation process [47]. However, the main chemical component of CRM is a blend of

synthetic and natural rubber with various other components, including black carbon, polymers, sulphur, paraffin, oil, fabrics, pigments and bead or belt materials [48]. Chemical modification of CRMB changes the typical chemical bonding between the rubber particles and bitumen [51, 52]. The FTIR measurements from a previous study showed that the addition of CR modifiers into control asphalt binder could delay and weaken the aging and oxidation effects. CRMB also improved the binder's permanent deformation and thermal cracking resistance [51].

#### 4.1.3. Mechanical Properties

The mechanical properties are based on the tensile strength ratio (TSR) and indirect tensile strength (ITS). The variation in shred size causes difficulty finding a considerable enough apparatus to carry out a significant shear test. A cohesion value of between 4.3 kPa to 11.5 kPa indicates that the tire chips have a permeability coefficient of between 1.5 to 15 ern/sec [52]. Tire chips have high insulating properties. The use of tire chips as fill material in subgrade applications results in a reduced depth of frost penetration relative to granular soil [32, 53, 54]. The dry process makes use of rubber as an aggregate. Typically, 2% to 3% rubber is prepended with the aggregates before charging the mixture with a pure asphalt binder [38]. However, the impact is mainly binder based, rubber content increases; the ITS value decreases.

#### 4.2. Crushed Concrete

Recycled crushed concrete is a suitable material for incorporation into HMA used to construct flexible road pavements. The use of recycled crushed concrete requires removing, breaking and crushing the concrete into a material with specific quality and size. Hence, there is a new generation of planting concrete (PC) made from recycled aggregates of demolished concrete to improve the fertiliser retention features and enable cost reduction. There are many benefits of using crushed concrete, and one of them is enhanced performance of pavement construction. The process of producing new road materials by crushing old roads and using the materials as the base layer for the new asphalt pavement is known as rubblisation [55-58].

#### **4.2.1.** Physical Properties

Research have shown that the incorporation of crushed brick and crushed concrete has a small to minimal effect on the physical and mechanical properties of the original material [8, 56, 59]. Up to 15% of crushed brick can be safely added to enhance the physical properties and improve pavement performance. RCA is characterised by low gravity and high-water absorption compared to natural aggregates [60], and this has the effect of enhancing the physical properties of the mixture. Based on the result of penetration index and from another side, it has reduced using the natural aggregate in the asphalt mixtures.

| Test Item                         | Test Result | Technical Requirement |
|-----------------------------------|-------------|-----------------------|
| Penetration (25°C, 100g, 5s)      | 53          | 40-60                 |
| Penetration index (PI)            | 0.06        | $\geq 0$              |
| Softening Point (°C)              | 87.0        | ≥ 75                  |
| Ductility at 5 °C (cm)            | 33.1        | ≥25                   |
| Dynamic viscosity at 135 °C (Pas) | 2.99        | ≤ 3                   |
| Flashpoint (°C)                   | 246         | ≥ 230                 |
|                                   |             |                       |

Table 3. The Physical Properties of Crushed Concrete modifier asphalt [56]

#### **4.2.2.** Mechanical Properties

The mechanical properties of the asphalt mix affected with a recycled concrete additive for the low-volume road classify them according to their applications in different pavement layers for different traffic volume roads. It is recommended that a certain percentage of RC additive in HMA be used for low-volume roads and useful for creating low-volume methods, as explained by the statistical model [61, 62]. The use of recycled concrete from damaged concrete can enhance the efficiency of asphalt mixtures, reduce the initial cost of production, and the volume of recyclable materials in the environment [64, 65]. In summary, compared to asphalt mixtures containing only natural aggregates, the mixtures containing recycled concrete RC additive have superior deformation strength, IDT strength, TSR, and rut depth properties. Also affected when the recycled concrete additive is added, which has enhanced the optimal bitumen content of the asphalt mixtures. Although the elastic modulus of the asphalt mixes was reduced, furthermore, the asphalt mixture modified with recycled concrete RC additive achieved lower stripping resistance than RC-free mixes [65].

#### 4.3. Plastic Waste

Waste polyethene terephthalate (PET) makes up about 60% of the plastic waste and is the primary component of plastic waste. The addition of PET as a modifier in asphalt binder and asphaltic mixtures besides treating aggregates against moisture and can increase the mixture's strength [66-69]. Thus, incorporation of plastic wastes into asphalt mixtures perform a dual function of enhancing the flexibility of pavement performance and reducing environmental pollution [70].

Many types of plastic recycling are derivatives from PET of plastic waste rich in polyethene, polyvinyl chloride, polystyrene, and polypropylene-like bottles, cups, and bags are used as additives in asphaltic mixtures. The recycled plastic does not release any toxic gas during the softening process [71].

Adjusted binder demonstrated the tremendous improvement in rheological characteristics to further binder materials, known as shredded plastic and high-density polyethene HDPE. The addition helps to solidify the mix.

#### 4.3.1. Chemical Properties

The chemical properties of asphalt modified binders were investigated using Fourier transform infrared spectroscopy (FTIR) and differential scanning calorimetry (DSC). These two methods complement each other [72]. Some mathematical equations showed significant correlations between the FTIR parameters and XRD parameters with PET parameters. Higher PET content increases aromaticity while reducing the oxygen-containing groups and aliphatic. Highly substituted aromatic rings are the primary aromatic features in low-rank bituminous PET, while condensed aromatic nuclei exist in the high-rank of bituminous PET and anthracite [59].

However, based on surface energy test to study the surface tension of the asphalt modifier, the functions of highdensity polyethene (HDPE) and low-density polyethene (LDPE) will impact the composite surface. Hence, the contact angle was measured to prove the surface energy and changes in the composites. The dispersive contributions and polar of the surface energies were calculated for the PET component created upon the initial contact angle values [73].

#### 4.3.2. Physical Properties

The physical properties of plastic waste as Polypropylene (PP) Plastic was used as a modifier for enhancement asphalt pavement durability. The result was very environmentally friendly [74], and the use of this material produced superior asphalt pavement relative to cement concrete pavement [75]. The incorporation of higher amounts of this plastic waste increased the softening point, penetration grade, and viscosity of the asphalt, and thus enhanced asphalt performance and reduced the effect of rutting [76, 77]. The mechanical performance of asphalt mixtures properties of PET is affected by crystallisation. Crystallinity is obtained by heating above the glass transition temperature; the chains become more flexible and can unfold under stress [78].

The rigidity of the binder increased as high as 4-6% of PET content due to the high compatibility between particles [79]. Two parameters that show the hardness, resistance to binder deformation under load, and fermentation or elasticity of asphalt, which are the complex shear and phase angle of the virgin binder, were drawn at different rates of LDPE, HDPE, against the temperature to deliver concurrent plots. The rutting border perimeter is plotted against temperature to obtain a master curve for the binders for investigation and analysis, as shown in Table 4 in some parts PP considered as the optimum in physical properties among LDPE and HDPE for the reason of PP chemical chain [80].

| Physical property               | Base bitumen | LDPE                          | HDPE                          | рр    |
|---------------------------------|--------------|-------------------------------|-------------------------------|-------|
| Specific Gravity                | 1.019        | Density=922 kg/m <sup>3</sup> | Density=961 kg/m <sup>3</sup> | 910   |
| Penetration @ 25 °C, 0.1mm      | 59.1         | 95                            | 127                           | 89    |
| Softening point (°C)            | 49.45        | 57                            | 54                            | 53    |
| Flashpoint (°C)                 | 310          |                               |                               | 322   |
| Ductility (mm)                  | 126.5        | 134                           | 130                           | 111   |
| Viscosity @ 135 °C (cP)         | 460.35       | 530                           | 570                           | 580   |
| Super-Pave performance Grade PG | 64-10        | 64-10                         | 64-10                         | 64-10 |

Table 4. The Physical properties for LDPE, HDPE and PP [79]

#### 4.3.3. Mechanical Properties

The addition of plastic bottles to the asphalt mixture increases the performance of the asphalt mixture and promotes their reuse [81], and a comparison was made between the original asphalt mixture and polymer-modified mix. Varying percentages of waste PET was added for improving the fatigue and stiffness properties of asphalt mixtures at optimum asphalt contents at three levels of stress (450, 350, 250 kPa), at 20°C. Results showed that the temperature of plastic asphalt is higher than the unmodified asphalt, which increases the characteristics of the road mix [82].

However, stability improved and water damage resistance increased when plastic material in strip shape less thickness and that to increase the surface area for reinforcement asphaltic mixture with the amount of 7% by weight of total [83]. The properties of high-performance asphalt mixtures can be improved through the addition of various combination of plastic wastes, namely, Polypropylene (PP) plastic and polyethylene (low-density and high density) [84]. HDPE is superior than LDPE and less expensive [85], and is preferred over LDPE to improve the mechanical properties of asphalt mixtures. Modified bituminous mixtures contain packaging material that has properties like to retain stability, Marshall Stability, and ITS better than unmodified bituminous mixes [86]. Waste plastic, which was used as a stabilising additive in asphalt mixture, was subjected to performance tests, and results showed that the addition of 10% shredded plastic produced a durable pavement [87]. According to summary reports (Mashaan et al. (2019b)), the use of waste plastic as a modifier in asphalt mixtures is an economical and effective method for preserving the environment from being further polluted. The use of plastic waste also has the benefits of improving bitumen's temperature susceptibility and stiffness, enhancing the rutting and fatigue cracking resistance of asphalt pavement [88, 89].

#### 4.4. Glass Fibre

That fibre can oppose the additional snugness in asphalt because of an expanded heap of development, consequently enhancing weariness life by expanding split opposition and constant twisting, particularly at high-weight level [90].

The conceivable focal points of utilising filaments to fortify black-top clearing blends incorporate decreased weariness, warm and intelligent splitting, expanded administration life and financial advantages [91]. The types of fibres strengthening, to a great extent, limit the forming of longitudinal incisions. Figure 5 shows the glass fibre waste material before and after recycling. Glass fibres can be classified into three categories:

- i. Natural fibres (banana fibre, cellulose, coconçut, jute, lignine, sisal).
- ii. Synthetic fibres (aramid fibres polyester, polypropylene, etc.).
- iii. Mineral fibres (carbon, asbestos, basalt, glass, steel fibres etc.).



Figure 5. Glass fibre waste material before and after recycling [92]

## 4.4.1. Physical Properties

The expansion of fibre-glass influenced the properties of bituminous blends by an increment in the stream esteem and diminishing its dependability and, additionally, the spaces in the blend [90]. The filaments can confer physical changes to bituminous blends, for example, support and toughening [93]. Isolated filaments give a high surface territory for every unit weight and act much like filler materials, which mass the bitumen wiping out total keep running off amid development [94]. Investigation of physical properties suggest that different sizes of fibre-glass tend to reduce penetration while increasing hardness and softening point [95]. These fibre sizes are within the recommended fibre dimensions used in previous studies (Table 5).

| Author                   | Fibre                       | Length (mm) | % Used  |
|--------------------------|-----------------------------|-------------|---------|
| Park et al., 2015 [96]   | Polyvinyl alcohol           | 13          | 0.5-1.5 |
| Otuoze et al., 2015 [97] | Waste HDPP                  |             | 0.5     |
| Vadoo et al., 2015       | Polypropylene & Polyester   | 10 - 20     |         |
| Woo et al., 2015 [98]    | Carbon, Mineral & Cellulose | 0.5 - 6     | 1 - 2   |
| Xiong et al., 2015 [99]  | Polyester                   | 6           |         |
| Xue et al., 2016 [100]   | Mineral fibre               | 2 - 8       | 3 - 9   |
| Anurag et al.,2009 [101] | Waste Polyester             | 13          | 0.5     |
| Chen et al., 2009 [102]  | Polyacrylonitrile           | 5           | 0.3     |
| Tapkin, 2008 [103]       | Propylene                   | 10          | 1       |

| <b>m i i f</b> |       |                 |         |         |             | •          | •   | •      | 1 0  |            |      | • •       | e .     |      |              |   | • •     |
|----------------|-------|-----------------|---------|---------|-------------|------------|-----|--------|------|------------|------|-----------|---------|------|--------------|---|---------|
|                | - I h | o rocommondor   | docor   | to ond  | <b>A</b> 11 | moncione   | tor | minoro | 1 11 | hroc nco   | A 0. | c rom     | arcomon | * ^* | ocnh         | പ | mivtiir |
| 1 410 5.2.     |       | етесонниениес   | i uusay | чс анни |             | THE ISTORS |     |        |      | ILLES LISE | U 71 | S I CI II |         |      | - <b>ASU</b> |   |         |
|                |       | e i eeominenaee |         |         |             |            |     |        |      |            | -    |           |         |      |              |   |         |

Given that each modifier has its advantage, the use of a combination of asphalt modifications could produce a competent asphalt binder. The length of fibre glass utilised significantly affects the versatile figure execution asphalt binder and its physical properties [7, 104, 105] It is accepted to transport physical changes to the bituminous blends by fortification and stress [93]. For activity, blends with filaments demonstrated a slight increment in the ideal cover content of the control blend [106].

The degree of sand dispersion in a blend determines the quality of the resulting mixture [107]. Adding fibre is valuable since it will help create an increasingly adaptable blend and in this way be progressively impervious to breaking [108, 107].

The results of conventional test of the physical properties showed that the addition of glass fibre enhanced the hardness of the modified asphalt binder. In general, an investigation on the conventional properties noted that adding glass fibre had a significant influence on asphalt binder rather than the unmodified [110].

Penetration value indicates the stiffness of an asphalt binder and softening point when the asphalt binder begins to become fluid. Table 6 shows the physical properties of modified glass fibre [111, 112].

Equally important, an illustration of the table noted that the optimum viscosity domain for asphalt binder was not too high for compaction asphalt mixture because the viscosity values reduce gradually with an increase in the temperatures [111-113].

| Bitumen                                  | Penetration at 25 °C (0.1-mm) | Softening point (°C) | Viscosity at 135°C (Pa.s) |  |  |  |
|--|-------------------------------|----------------------|---------------------------|--|--|--|
| Fibre-glass modified binder              | 54                            | 59                   | 0.60                      |  |  |  |
| Rubber-modified binder with fibre glass  | 59                            | 55                   | 0.56                      |  |  |  |
| Polymer-modified binder with fibre glass | 48                            | 59                   | 0.62                      |  |  |  |

Table 6. The physical properties of glass fibre-modified asphalt [111]

#### 4.4.2. Chemical Properties

The chemical composition of a raw material determines which chemical propertied of asphalt is affected [114, 115]. Therefore, the interfering phenomena of the chains in the spectra are common in complex materials. The peaks in spectra present the infrared absorbance in the materials, and it also reflects the extent of bonds in the materials. Hence, if the quantity of bonds in the asphalt binder is too small, it means that the difficulty of revelation is growing. According to the surface composition, it is rich in rubber as confirmed by S from vulcanising and by Ca and Si utilised as oxides in fillers and HDPE. These aggregates are responsible for the crystalline part of the composite because of the HDPE content. Mechanical interfaces that were highlighted involved rubbers (with traces of Si, Zn, S) and HDPE. This finding has been proven by FTIR, which showed a null or slight presence of glass fibre, as this zone has deficient oxygen. The same result indicates the PET scrap, as recording a high percentage of oxygen and the amount of Na (sample 2a-CS) following glass fibre functionalization with SDS. Elemental analysis showed that the glass fibre functionalised composite involves large voids, particularly more abundant for the Type P composite. Hence, this ascertains that SDS promotes a better attraction among glass fibre and the hydrocarbon components, thus denser composites [73]. The addition of this modifier into a modified asphalt binder as a compatible modifier tends to disperse the asphalt matrix and create a homogeneous mixture that is susceptible to temperature with high-temperature storage stability [116].

#### 4.4.3. Mechanical Properties

The performance testing of asphalt mixes is conducted to evaluate the glass fibre and enhance the performance of pavements [111]. The modification of asphalt mixtures with glass fibre can improve the mechanical properties of asphalt mixtures, including the flow value, stability, and voids. Several studies have shown that modification of asphalt with 1.25 mass of low density of glass fibre resulted in enhanced fatigue rutting and rheological properties of the mixture [117, 118]. A study with different factors to evaluate the performance of fibres in asphalt mixture and their findings revealed that 4 - 6% of this material could be added to the asphalt mixture for enhancement of resistance to permanent deformation, increase the stiffness (Marshall Stability), and control flow of the mixture. Results also provide lower binder drain down and save the environment by reducing waste plastic environmental pollution [81].

#### 4.5. Steel Slag

Steel slags are the by-product from steel manufacturing [119]. There are two kinds of steel slag, namely basic oxygen furnace (BOF) and electric arc furnace (EAF) [120]. The type of steel slag is determined by the cooling down process and chemical composition of the slag [74, 121, 122]. In 2006, an estimated 114-280 million tons steel slag was produced globally [123]. In 2013, 37% of the steel slag was used in road pavement, and 97% of the steel slag produced in Germany is used as aggregates in the in the base, sub-base and surface layer of road pavements [124]. In Malaysia, steel slag is utilised as aggregate replacement for sustainable construction of highways. Steel slag aggregate benefits the environment and reduces the amount of granite used in the construction of highways [125]. It can be used to stabilise HMA. Pavement containing steel slag mixed with asphalt is better than ordinary asphalt pavement and hence utilised in road pavement, for example, construction of expressway [121, 126].

#### **4.5.1.** Physical Properties

The features of steel slag, for example, substantial ratio, excellent abrasion resistance, rough surface and firm combination with asphalt can be utilised widely in road engineering. In the same view, the aggregates of steel slag are flat shaped and too angular, with cells which are not well connected that it gives a firm surface area rather than the natural aggregates, and this steel slag aggregate provides a great bond with the bitumen [100, 127]. Table 7 shows the physical properties of steel slag.

| Property  | Value     |
|---|-----------|
| Los Angeles abrasion (ASTM C131), %                         | 20-25     |
| Sodium sulphate soundness loss ASTM C88), %                 | < 12      |
| Angle of internal friction                                  | 40-50     |
| Hardness (measured by Moh's scale of mineral hardness)1     | 6-7       |
| California bearing ratio (CBR), % top size 19mm (3/4 inch)2 | Up to 300 |

Table 7. The physical properties of steel slag [128]

#### 4.5.2. Chemical Properties

The chemical evolution of slag needs further concoction arrangement of steel slag, an unpredictable lattice structure comprising mostly of straightforward oxides decided from an essential examination of x-beam fluorescence. Minor components incorporate sulphur, press, manganese, soluble bases and a few others. Table 8 presents the rundown of different scopes of mixes contained in steel slag as revealed by Yildirim and Prezzi (2011) [129].

| Constituent                                       | Open hearth steel slag | Basic oxygen |
|---|------------------------|--------------|
| Calcium oxide (CaO)                               | 258.8                  | 41.3         |
| Silicon dioxide (SiO <sub>2</sub> )               | 16.4                   | 15.6         |
| Iron (FeO or Fe <sub>2</sub> O <sub>3</sub> )     | 26.0                   | 20.0         |
| Magnesium oxide (MgO)                             | 10.0                   | 6.9          |
| Manganese oxide (MnO)                             | 11.2                   | 8.9          |
| Aluminium oxide (Al <sub>2</sub> O <sub>3</sub> ) | 2.4                    | 2.2          |
| Titanium dioxide (TiO <sub>2</sub> )              | 0.8                    | 0.5          |
| Free lime (free CaO)                              | 2.1                    | 3.3          |
|   |                        |              |

Liu et al. (2019) evaluated the chemical composition of the steel slag using X-ray fluorescence spectrometer (XRF), and the mineral composition was examined and analysed using X-ray diffractometer (XRD). The asphalt was analysed by Fourier Transform Infrared Spectroscopy (FTIR). The primary minerals present in steel slag are Ferro ferric-oxide Fe<sub>2</sub>O<sub>3</sub>, i.e., dicalcium silicate, tricalcium silicate and triton tetroxide, which chemically react with asphalt to create new substances [130].

#### 4.5.3. Mechanical Properties

According to The Federal Highway Administration (FHWA), steel slag has useful mechanical properties that can markedly improve the performance of asphalt pavement [131]. Economists recommend adding SSA to the asphalt mixture for road constructions to reduce the cost of extracting and processing natural aggregates as well as the cost of transporting and treating steel waste at vast number of stockpiles [28, 132]. The use of natural aggregates in the asphalt layers of road pavement is not cost-effective when steel slag is readily available. The use of waste materials is very beneficial for the environment, economy, and society [133, 134].

### 5. Challenges in Incorporating Waste Materials in Asphalt Pavements

The use of waste materials in road and pavement construction and throughout the globe has many advantages; among the benefits are the low initial cost to develop infrastructure and that requires advanced technology to prepare and process to be implementable to meet the requirements of pavement materials [5]. Besides, existences waste materials affecting the high costs of pavement construction. However, decrease the cost of pavement construction in the long-term. The potential difficulties and barriers in using waste materials in the pavement sector are related to the recycling process. Some sensors detect and sort waste based on chemical composition, the durability of the resulting asphalt and concrete mix can also be affected [41]. The private sector could adopt a possible solution way to such a logistical case. The recycling of infrastructure development based on the public-private partnership sector. Governments can offer incentives in the form of tax exemption might to encourage pavement technologists and contractors to use waste materials. This policy of sustainable production has been successfully implemented in China [53].

Moreover, training and facilitating paving crews and material technologists to use a different type of waste materials pavement construction and practice codes should be provided. Currently, there is no standardised practice code using waste glass, plastic, or steel slag. The appropriate number of uses depends on the type of waste material. It is impossible to define an answer that encompasses every composition. Furthermore, various waste materials must be ranked based on their potential for recyclability.

## 6. Considerations of Environmental and Economic Aspect

The use of recycled waste materials in asphalt mixture and asphalt binder is a method for managing waste material sustainably and to produce a superior binder for constructing asphalt mixture [135, 136]. Waste materials have promising potential in terms of economic and environmental benefits. Both developing and developed countries demolish buildings on a daily basis [2]. Countries that are densely populated and have scarce liveable land areas have to minimise landfill areas. The use of demolition waste in an asphaltic mixture can reduce the need for landfilling. On the other hand, tire rubber can be used in asphaltic mixture and asphalt binder [79]. This waste provides a low-cost solution both as an additive and binder modifier.

Different types of plastic waste, glass fibre and steel sludge can be recycled in the asphaltic mixture and contribute to sustainable practice and becomes an industry practice for constructing stone mastic asphalt and can be effectively used as a filler in asphalt mixtures [15]. The decision to use industrial waste materials in concrete pavements reflects environmental pollution, maintenance liability, technological risk, resources provided and cost of production. There are still a few problems faced by many researchers about leaching. For instance, the crumb rubber modified asphalt contains heavy metals such as polycyclic aromatic hydrocarbons (PAHs) and zinc (Zn) due to the high concentration of toxic chemicals that can affect our health and the environment [5]. The result of heavy metal analysis showed that heavy metals in steel slag do not pose any environmental risks or leaching problems when used as aggregates in road applications [137]. Besides, the environmental impact of waste materials has raised concerns about the possible environmental harm caused by the improper handling of recycled materials, as well as concerns about leaks, spills, dumps, or other forms of hazardous substances released into the environment.

## 7. Conclusions

This study has conducted a feasibility test on the alternative materials used in asphalt mixture and transportation infrastructure. These alternative materials, for instance, plastic waste, crumb rubber, glass fibre, steel slag and crushed concrete, showed promising results as they could be used as an alternative aggregates substance that could be used in the asphalt mixture design. These alternatives not necessarily be an aggregate replacement alternative but could be an

additional substance to the aggregate to increase particular physical and chemical attributes from the asphalt mix design. Although there is research on the potential use of waste material in road pavements, there are knowledge gaps and concerns that need concentrated study and assessment in the interest of building better paving and preserving natural resources. There are many advantages in using waste material in asphalt mixtures as it is useful for environmental issues. The following conclusions are drawn and should be taken into consideration by future studies.

- Waste material can be utilised as a choice aggregate in asphalt mixtures. Nevertheless, the primary chemical properties and impurities must be taken into consideration.
- Waste materials are given a new chapter of life as a low-cost novel additive to asphalt mixture. Waste materials can give superior performance to asphalt binders and promote cleaner and sustainable production.
- The use of waste plastic as a recycled polymer to modify asphalt could produce the same result as commercially manufactured polymers by increasing rutting resistance, fatigue life, and reducing cost. The reuse of plastic waste can contribute considerably to protecting the environment from future pollution.
- All the studies on waste material revealed almost different possible ratios from others. Hence, the development of fixed replacement ratio is required. The reason was that the features of the waste materials depend on various factors, for example, quality, source of extraction, handling, variation in the composition of minerals etc.
- The problem of quality control: Wastes have a good potential for use as a replacement for conventional material in flexible pavement, but it is not feasible to use waste materials as they are made up of several undesirable mixtures of materials.

## 8. Funding and Acknowledgement

The authors would like to express their gratitude to Universiti Kebangsaan Malaysia for the financial support for this work through Grant No. GUP-2018-094.

## 9. Conflicts of Interest

The authors declare no conflict of interest.

## **10. References**

- Jamshidi, Ali, and Greg White. "Evaluation of Performance and Challenges of Use of Waste Materials in Pavement Construction: A Critical Review." Applied Sciences 10, no. 1 (2020): 226. doi:10.3390/app10010226.
- [2] Milad, Abdalrhman, Aizat Mohd Taib, Abobaker GF Ahmeda, Mohmed Solla, and Nur Izzi Md Yusoff. "A Review of the Use of Reclaimed Asphalt Pavement for Road Paving Applications." Jurnal Teknologi 82, no. 3 (2020). doi:10.11113/jt.v82.14320.
- [3] F. A. Santagata, F. Canestrari, and E. Pasquini, 'Mechanical characterisation of asphalt rubber-wet process', in Palermo: sn In: 4th International SIIV Congress, 2007.
- [4] Abdul-Raouf, Manar E., Nermine E. Maysour, Abdul-Azim A. Abdul-Azim, and Mahasen S. Amin. "Thermochemical recycling of mixture of scrap tyres and waste lubricating oil into high caloric value products." Energy conversion and management 51, no. 6 (2010): 1304-1310. doi:10.1016/j.enconman.2010.01.007.
- [5] Milad, Abdalrhman, Abobaker GF Ahmeda, Aizat Mohd Taib, Suzielah Rahmad, Mohmed Solla, and Nur Izzi Md Yusoff. "A review of the feasibility of using crumb rubber derived from end-of-life tire as asphalt binder modifier." Journal of Rubber Research (2020): 1-14. doi:10.1007/s42464-020-00050-y.
- [6] Choudhary, Jayvant, Brind Kumar, and Ankit Gupta. "Utilisation of solid waste materials as alternative fillers in asphalt mixes: A review." Construction and Building Materials 234 (2020): 117271. doi:10.1016/j.conbuildmat.2019.117271.
- [7] Sunthonpagasit, Nongnard, and Michael R. Duffey. "Scrap tires to crumb rubber: feasibility analysis for processing facilities." Resources, Conservation and recycling 40, no. 4 (2004): 281-299. doi:10.1016/S0921-3449(03)00073-9.
- [8] Adhikari, B., Debapriya De, and S. Maiti. "Reclamation and recycling of waste rubber." Progress in polymer science 25, no. 7 (2000): 909-948. doi:10.1016/S0079-6700(00)00020-4.
- [9] de Almeida Júnior, Adão Francisco, Rosane Aparecida Battistelle, Barbara Stolte Bezerra, and Rosani de Castro. "Use of scrap tire rubber in place of SBS in modified asphalt as an environmentally correct alternative for Brazil." Journal of Cleaner Production 33 (2012): 236-238. doi:10.1016/j.jclepro.2012.03.039.
- [10] Jamal, Muhammad, and Filippo Giustozzi. "Low-content crumb rubber modified bitumen for improving Australian local roads condition." Journal of Cleaner Production 271 (2020): 122484. doi:10.1016/j.jclepro.2020.122484.
- [11] Townsend, Stephen W., Chad J. Spreadbury, Steven J. Laux, Christopher C. Ferraro, Ramana Kari, and Timothy G. Townsend. "Blending as a Strategy for Reusing Municipal Solid Waste Incinerator Ash in Road-Base Construction." Journal of Environmental Engineering 146, no. 9 (2020): 04020106. doi:10.1061/(ASCE)EE.1943-7870.0001788.

- [12] Debbarma, Solomon, G. D. Ransinchung, Surender Singh, and Surya Kant Sahdeo. "Utilisation of industrial and agricultural wastes for productions of sustainable roller compacted concrete pavement mixes containing reclaimed asphalt pavement aggregates." Resources, Conservation and Recycling 152 (2020): 104504. doi:10.1016/j.resconrec.2019.104504.
- [13] Nwakaire, Chidozie Maduabuchukwu, Soon Poh Yap, Chiu Chuen Onn, Choon Wah Yuen, and Hussein Adebayo Ibrahim. "Utilisation of recycled concrete aggregates for sustainable highway pavement applications; a review." Construction and Building Materials 235 (2020): 117444. doi:10.1016/j.conbuildmat.2019.117444.
- [14] Zou, Guilian, Xuekai Sun, Xinhai Liu, and Junjie Zhang. "Influence factors on using recycled concrete aggregate in foamed asphalt mixtures based on tensile strength and moisture resistance." Construction and Building Materials 265 (2020): 120363. doi:10.1016/j.conbuildmat.2020.120363.
- [15] Environment Protection Authority, Standard for the production and use of waste derived soil enhancer. 2010.
- [16] M. I. Souliman, R. A. El-Hakim, M. Davis, H. Gc, and L. Walubita, 'Mechanistic and Economic Impacts of Using Asphalt Rubber Mixtures at Various Vehicle Speeds', Advances in Civil Engineering Materials, vol. 7, no. 3, pp. 347–359, 2018. doi:10.1520/ACEM20170104.
- [17] Mashaan, Nuha Salim, Asim H. Ali, Suhana Koting, and Mohamed Rehan Karim. "Performance evaluation of crumb rubber modified stone mastic asphalt pavement in Malaysia." Advances in Materials Science and Engineering 2013 (2013).. doi:10.1155/2013/304676.
- [18] Arabani, Mahyar, Seyed Amid Tahami, and Gholam Hossein Hamedi. "Performance evaluation of dry process crumb rubbermodified asphalt mixtures with nanomaterial." Road Materials and Pavement Design 19, no. 5 (2018): 1241-1258. doi:10.1080/14680629.2017.1302356.
- [19] Pasandín, A. R., and I. Pérez. "Fatigue performance of bituminous mixtures made with recycled concrete aggregates and waste tire rubber." Construction and Building Materials 157 (2017): 26-33. doi:10.1016/j.conbuildmat.2017.09.090.
- [20] Ahmadinia, Esmaeil, Majid Zargar, Mohamed Rehan Karim, Mahrez Abdelaziz, and Ebrahim Ahmadinia. "Performance evaluation of utilisation of waste Polyethylene Terephthalate (PET) in stone mastic asphalt." Construction and Building Materials 36 (2012): 984-989. doi:10.1016/j.conbuildmat.2012.06.015.
- [21] Ziari, Hassan, Arman Ghasemi Kaliji, and Rezvan Babagoli. "Laboratory evaluation of the effect of waste plastic bottle (PET) on rutting performance of hot mix asphalt mixtures." Petroleum Science and Technology 34, no. 9 (2016): 819-823. doi:10.1080/10916466.2016.1169290.
- [22] G. White, 'Evaluating recycled waste plastic modification and extension of bituminous binder for asphalt', in Eighteenth Annual International Conference on Pavement Engineering, Asphalt Technology and Infrastructure, Liverpool, England, United Kingdom, 2019.
- [23] Fu, Zhen, Wanqing Shen, Yue Huang, Guolin Hang, and Xiaotong Li. "Laboratory evaluation of pavement performance using modified asphalt mixture with a new composite reinforcing material." International Journal of Pavement Research and Technology 10, no. 6 (2017): 507-516. doi:10.1016/j.ijprt.2017.04.001.
- [24] Celauro, Clara, and F. G. Praticò. "Asphalt mixtures modified with basalt fibres for surface courses." Construction and Building Materials 170 (2018): 245-253. doi:10.1016/j.conbuildmat.2018.03.058.
- [25] Wu, Mengmeng, Rui Li, Yuzhen Zhang, Jianming Wei, Yuchao Lv, and Xuemei Ding. "Reinforcement effect of fiber and deoiled asphalt on high viscosity rubber/SBS modified asphalt mortar." Petroleum Science 11, no. 3 (2014): 454-459. doi:10.1007/s12182-014-0361-2.
- [26] Maharaj, Chris, Daniel White, Rean Maharaj, and Cheryon Morin. "Reuse of steel slag as an aggregate to asphaltic road pavement surface." Cogent Engineering 4, no. 1 (2017): 1416889. doi:10.1080/23311916.2017.1416889.
- [27] Gowtham, C., and K. Ganesh. "Effect of Steel Slag on Marshall Properties of Plain and Modified Bituminous Concrete Grade 1 Mix." International Journal of Engineering Research & Technology (IJERT) (2018): 2278-0181.
- [28] Li, Qiushi, Yanjun Qiu, Ali Rahman, and Haibo Ding. "Application of steel slag powder to enhance the low-temperature fracture properties of asphalt mastic and its corresponding mechanism." Journal of Cleaner Production 184 (2018): 21-31. doi:10.1016/j.jclepro.2018.02.245.
- [29] Huang, Yue, Roger N. Bird, and Oliver Heidrich. "A review of the use of recycled solid waste materials in asphalt pavements." Resources, conservation and recycling 52, no. 1 (2007): 58-73. doi:10.1016/j.resconrec.2007.02.002.
- [30] Kaya, Muammer. "Recovery of metals and nonmetals from electronic waste by physical and chemical recycling processes." Waste management 57 (2016): 64-90. doi:10.1016/j.wasman.2016.08.004.
- [31] Menaria, Yash, and Rupal Sankhla. "Use of waste plastic in flexible pavements-green roads." Open Journal of Civil Engineering 5, no. 03 (2015): 299.

- [32] Mohajerani, Abbas, Lucas Burnett, John V. Smith, Stefan Markovski, Glen Rodwell, Md Tareq Rahman, Halenur Kurmus et al. "Recycling waste rubber tyres in construction materials and associated environmental considerations: A review." Resources, Conservation and Recycling 155 (2020): 104679. doi:10.1016/j.resconrec.2020.104679.
- [33] Chesner, Warren H., Robert J. Collins, Michael H. MacKay, and John Emery. User guidelines for waste and by-product materials in pavement construction. No. FHWA-RD-97-148, Guideline Manual, Rept No. 480017. Recycled Materials Resource Center, 2002.
- [34] Lewandowski, L. H. "Polymer modification of paving asphalt binders." Rubber Chemistry and Technology 67, no. 3 (1994): 447-480.
- [35] Sofi, Alba. "Effect of waste tyre rubber on mechanical and durability properties of concrete-A review." Ain Shams Engineering Journal 9, no. 4 (2018): 2691-2700. doi:10.1016/j.asej.2017.08.007.
- [36] Bernier, Alexander, Adam Zofka, and Iliya Yut. "Laboratory evaluation of rutting susceptibility of polymer-modified asphalt mixtures containing recycled pavements." Construction and Building Materials 31 (2012): 58-66. doi:10.1016/j.conbuildmat.2011.12.094.
- [37] Alsaleh, Ali, and Melanie L. Sattler. "Waste tire pyrolysis: influential parameters and product properties." Current Sustainable/Renewable Energy Reports 1, no. 4 (2014): 129-135.
- [38] Takallou, H. Barry, Hussain U. Bahia, Dario Perdomo, and Robert Schwartz. "Use of Superpave technology for design and construction of rubberised asphalt mixtures." Transportation Research Record 1583, no. 1 (1997): 71-81.
- [39] Heitzman, Charles J. "Shower head with continuous or cycling flow rate, fast or slow pulsation and variable spray pattern." U.S. Patent 6,223,998, issued May 1, 2001.
- [40] Fukumori, Kenzo, and Mitsumasa Matsushita. "Material recycling technology of crosslinked rubber waste." R&D Review of Toyota CRDL 38, no. 1 (2003): 39-47.
- [41] Ibrahim, Mohd Rasdan, Herda Yati Katman, Mohamed Rehan Karim, Suhana Koting, and Nuha S. Mashaan. "A review on the effect of crumb rubber addition to the rheology of crumb rubber modified bitumen." Advances in Materials Science and Engineering 2013 (2013).
- [42] Kaushik, M. K., Arvind Kumar, and Ajay Bansal. "Geo-Environmental prospective and development plans for a new MSW landfill site using tire chips as leachate drainage material." Electron J Geotech Eng 19, no. Z6 (2014): 17495-17513.
- [43] Zhang, Jizhe, Hongyin Li, Peng Liu, Ming Liang, Hongguang Jiang, Zhanyong Yao, and Gordon Airey. "Experimental Exploration of Influence of Recycled Polymer Components on Rutting Resistance and Fatigue Behavior of Asphalt Mixtures." Journal of Materials in Civil Engineering 32, no. 6 (2020): 04020129.
- [44] Chirila, Elisabeta, and Camelia Draghici. "Risk assessment of mixtures of chemical pollutants in the environment." In Exposure and Risk Assessment of Chemical Pollution—Contemporary Methodology, pp. 51-67. Springer, Dordrecht, 2009. doi:10.1007/978-90-481-2335-3\_4.
- [45] Xiao, Feipeng, Serji Amirkhanian, and C. Hsein Juang. "Rutting resistance of rubberised asphalt concrete pavements containing reclaimed asphalt pavement mixtures." Journal of Materials in Civil Engineering 19, no. 6 (2007): 475-483. doi:10.1061/(ASCE)0899-1561(2007)19:6(475).
- [46] Yao, Hui, Zhanping You, Liang Li, Shu Wei Goh, Chee Huei Lee, Yoke Khin Yap, and Xianming Shi. "Rheological properties and chemical analysis of nanoclay and carbon microfiber modified asphalt with Fourier transform infrared spectroscopy." Construction and Building Materials 38 (2013): 327-337. doi:10.1016/j.conbuildmat.2012.08.004.
- [47] Rocha, Magda, Alexandra Mansur, and Herman Mansur. "Characterisation and accelerated ageing of UHMWPE used in orthopedic prosthesis by peroxide." Materials 2, no. 2 (2009): 562-576. doi:10.3390/ma2020562.
- [48] Council, Auckland Regional. Literature Review of Organic Chemicals of Emerging Environmental Concern in Use in Auckland. No. 028. 1st edi. Technical Report, 2008.
- [49] Wang, Haopeng, Xueyan Liu, Panos Apostolidis, Martin van de Ven, Sandra Erkens, and Athanasios Skarpas. "Effect of laboratory aging on chemistry and rheology of crumb rubber modified bitumen." Materials and Structures 53, no. 2 (2020): 1-15.
- [50] Liu, HuiLin, XiaoPing Wang, and DeMin Jia. "Recycling of waste rubber powder by mechano-chemical modification." Journal of Cleaner Production 245 (2020): 118716. doi:10.1016/j.jclepro.2019.118716.
- [51] Rodríguez-Alloza, Ana María, Juan Gallego, and Felice Giuliani. "Complex shear modulus and phase angle of crumb rubber modified binders containing organic warm mix asphalt additives." Materials and Structures 50, no. 1 (2017): 77. doi:10.1617/s11527-016-0950-1.

- [52] Azmi, Ahmad Azrem, Mohd Mustafa Abdullah Al Bakri, Che Mohd Ruzaidi Ghazali, Andrei Victor Sandu, Kamarudin Hussin, and Donny Arief Sumarto. "A Review on Fly Ash Based Geopolymer Rubberized Concrete." In Key Engineering Materials, vol. 700, pp. 183-196. Trans Tech Publications Ltd, 2016. doi:10.4028/www.scientific.net/KEM.700.183.
- [53] Usman, Nura, Mohd Idrus Mohd Masirin, Kabiru Abdullahi Ahmad, and Ahmad Suliman B. Ali. "Application of Recycled Polyethylene Terephthalate Fiber in Asphaltic Mix for Fatigue Life Improvement." In Global Civil Engineering Conference, pp. 1401-1413. Springer, Singapore, 2017. doi:10.1007/978-981-10-8016-6\_101.
- [54] Movilla-Quesada, D., A. C. Raposeiras, and J. Olavarría. "Effects of Recycled Polyethylene Terephthalate (PET) on Stiffness of Hot Asphalt Mixtures." Advances in Civil Engineering 2019 (2019). doi:10.1155/2019/6969826.
- [55] Jung, Simon, Sebastian Peetz, and Michael Koch. "POEAM A Method for the Part Orientation Evaluation for Additive Manufacturing." HighTech and Innovation Journal 1, no. 1 (March 1, 2020): 21–27. doi:10.28991/hij-2020-01-01-03.
- [56] Aatheesan, T., Arul Arulrajah, G. Newman, Myint Win Bo, and J. Wilson. "Crushed brick blends with crushed concrete for pavement sub-base and drainage applications." Australian Geomechanics 44, no. 2 (2009): 65. doi:10.1139/t2012-041.
- [57] Viyanant, Chirayus. "Potential use of recycled asphalt pavement and crushed concrete as backfill for mechanically stabilised earth walls." PhD diss., 2006.
- [58] Milad, Abdalrhman, Noor Ezlin Ahmed Basri, Mohammad K. Younes, Hassan M. Abdelsalam, and Riza Atiq Abdullah Bin OK Rahmat. "Selecting the Affected Factors on Pavement Distress Problems Using Analytical Hierarchy Process [AHP]." International Journal of Engineering & Technology 7, no. 2.29 (2018): 716-718. doi:10.14419/ijet.v7i2.29.14004.
- [59] Verian, Kho Pin, Nancy M. Whiting, Jan Olek, Jitendra Jain, and Mark B. Snyder. "Using recycled concrete as aggregate in concrete pavements to reduce materials cost." (2013). doi:10.5703/1288284315220.
- [60] Xiao, Y., M. F. C. van de Ven, A. A. A. Molenaar, and S. P. Wu. "Possibility of using epoxy modified bitumen to replace tarcontaining binder for pavement antiskid surfaces." Construction and Building Materials 48 (2013): 59-66. doi:10.1016/j.conbuildmat.2013.06.080.
- [61] Mills-Beale, Julian, and Zhanping You. "The mechanical properties of asphalt mixtures with recycled concrete aggregates." Construction and Building Materials 24, no. 3 (2010): 230-235. doi:10.1016/j.conbuildmat.2009.08.046.
- [62] Bhusal, Sushanta, Xiaojun Li, and Haifang Wen. "Evaluation of effects of recycled concrete aggregate on volumetrics of hotmix asphalt." Transportation research record 2205, no. 1 (2011): 36-39. doi:10.3141/2205-05.
- [63] Dehnad, M. H., A. Khodaii, and F. Moghadas Nejad. "Moisture sensitivity of asphalt mixtures under different load frequencies and temperatures." Construction and Building Materials 48 (2013): 700-707. doi:10.1016/j.conbuildmat.2013.07.059.
- [64] Tahmoorian, Farzaneh, and Bijan Samali. "Laboratory investigations on the utilisation of RCA in asphalt mixtures." International Journal of Pavement Research and Technology 11, no. 6 (2018): 627-638. doi:10.1016/j.ijprt.2018.05.002.
- [65] Qasrawi, Hisham, and Ibrahim Asi. "Effect of bitumen grade on hot asphalt mixes properties prepared using recycled coarse concrete aggregate." Construction and Building Materials 121 (2016): 18-24. doi:10.1016/j.conbuildmat.2016.05.101.
- [66] Rahman, Wan Mohd Nazmi Wan Abdul, and Achmad Fauzi Abdul Wahab. "Green pavement using recycled polyethylene terephthalate (PET) as partial fine aggregate replacement in modified asphalt." Procedia Engineering 53 (2013): 124-128. doi:10.1016/j.proeng.2013.02.018.
- [67] Araujo Scharnberg, Allan Ramone, Adrison Carvalho de Loreto, and Annelise Kopp Alves. "Optical and Structural Characterization of Bi2FexNbO7 Nanoparticles for Environmental Applications." Emerging Science Journal 4, no. 1 (February 1, 2020): 11–17. doi:10.28991/esj-2020-01205.
- [68] Iqbal, Maria, Arshad Hussain, Afaq Khattak, and Kamran Ahmad. "Improving the Aging Resistance of Asphalt by Addition of Polyethylene and Sulphur." Civil Engineering Journal 6, no. 5 (2020): 1017-1030. doi:10.28991/cej-2020-03091525.
- [69] Omrani, Hamed, Ali Reza Ghanizadeh, and Amin Tanakizadeh. "Effect of SBS Polymer and Anti-stripping Agents on the Moisture Susceptibility of Hot and Warm Mix Asphalt Mixtures." Civil Engineering Journal 3, no. 10 (2017): 987-996. doi:10.28991/cej-030931.
- [70] Sojobi, Adebayo Olatunbosun, Stephen Emeka Nwobodo, and Oluwasegun James Aladegboye. "Recycling of polyethylene terephthalate (PET) plastic bottle wastes in bituminous asphaltic concrete." Cogent engineering 3, no. 1 (2016): 1133480. doi:10.1080/23311916.2015.1133480.
- [71] S Karmakar, Sandip, and Tapas Kumar Roy. "Effect of waste plastic and waste tires ash on mechanical behavior of bitumen." Journal of Materials in Civil Engineering 28, no. 6 (2016): 04016006. doi:10.1061/(asce)mt.1943-5533.0001484.
- [72] Fan, Zhenyang, Xuancang Wang, Zhuo Zhang, and Yi Zhang. "Effects of cement-mineral filler on asphalt mixture performance under different aging procedures." Applied Sciences 9, no. 18 (2019): 3785. doi:10.3390/app9183785.

- [73] Cazan, Cristina, Mihaela Cosnita, and Anca Duta. "Effect of PET functionalisation in composites of rubber–PET–HDPE type." Arabian Journal of Chemistry 10, no. 3 (2017): 300-312. doi:10.1016/j.arabjc.2015.10.005.
- [74] Masirin, Mohd Idrus Mohd, Ahmed Suliman B. Ali, Mohammad Sukri Mustapa, Raha Abdul Rahman, Abdullah Wagiman, and Muhammad Isom Aziz. "Analysis of Physical and Microstructural Properties on Parit Nipah Peat Particles as Sustainable Asphalt Modifier." Materials Science Forum 975 (January 2020): 197–202. doi:10.4028/www.scientific.net/msf.975.197.
- [75] Wu, Shenghua, and Luke Montalvo. "Repurposing Waste Plastics into Cleaner Asphalt Pavement Materials: A Critical Literature Review." Journal of Cleaner Production 280 (January 2021): 124355. doi:10.1016/j.jclepro.2020.124355.
- [76] Abdullah, Mohd Ezree, Nurul Asyiqin Ahmad, Ramadhansyah Putra Jaya, Norhidayah Abdul Hassan, Haryati Yaacob, and Mohd Rosli Hainin. "Effects of Waste Plastic on the Physical and Rheological Properties of Bitumen." IOP Conference Series: Materials Science and Engineering 204 (May 2017): 012016. doi:10.1088/1757-899x/204/1/012016.
- [77] Vasudevan, R., A. Ramalinga Chandra Sekar, B. Sundarakannan, and R. Velkennedy. "A technique to dispose waste plastics in an ecofriendly way–Application in construction of flexible pavements." Construction and Building Materials 28, no. 1 (2012): 311-320. doi:10.1016/j.conbuildmat.2011.08.031.
- [78] Rasool, Dalia Adil, Besma Mohammed Fahad, and Khalid Mershed Awaeed. "Utilization of Waste Plastic Water Bottle as a Modifier For Asphalt mixture Properties." Journal of Engineering and Sustainable Development 19, no. 2 (2015): 89-108.
- [79] Rahman, Md Tareq, Abbas Mohajerani, and Filippo Giustozzi. "Recycling of Waste Materials for Asphalt Concrete and Bitumen: A Review." Materials 13, no. 7 (2020): 1495. doi:10.3390/ma13071495.
- [80] Hassan, N.A., R. Khan, J. Raaberg, and D. Lo Presti. "Effect of Mixing Time on Reclaimed Asphalt Mixtures: An Investigation by Means of Imaging Techniques." Construction and Building Materials 99 (November 2015): 54–61. doi:10.1016/j.conbuildmat.2015.09.009.
- [81] Ahmadinia, Esmaeil, Majid Zargar, Mohamed Rehan Karim, Mahrez Abdelaziz, and Payam Shafigh. "Using waste plastic bottles as additive for stone mastic asphalt." Materials & Design 32, no. 10 (2011): 4844-4849. doi:10.1016/j.matdes.2011.06.016.
- [82] H. Mahfouz et al., 'Using Recycled Plastic as Hot Mix Asphalt Modifiers', Resilient Infrastructure, London, England, United Kingdom, (2016).
- [83] Jassim, Hamed M., Omar T. Mahmood, and Sheelan A. Ahmed. "Optimum use of plastic waste to enhance the Marshall properties and moisture resistance of hot mix asphalt." International Journal Engineering. Trends and Technology 7 (2014): 18-25. doi:10.14445/22315381/ijett-v7p223.
- [84] Costa, Liliana MB, H. M. R. D. Silva, Joel RM Oliveira, and Sara RM Fernandes. "Incorporation of waste plastic in asphalt binders to improve their performance in the pavement." International journal of pavement research and technology 6, no. 4 (2013): 457-464.
- [85] Choudhary, A. K., J. N. Jha, and K. S. Gill. "Utilisation of plastic wastes for improving the sub-grades in flexible pavements." In Paving materials and pavement analysis, pp. 320-326. 2010. doi:10.1061/41104(377)39.
- [86] Sangita, Gupta Reena, and Kaur Verinder. "A novel approach to improve road quality by utilizing plastic waste in road construction." Journal of Environmental Research and Development 5, no. 4 (2011): 1036-1042.
- [87] C. S. Bindu and K. S. Beena, 'Waste plastic as a stabilising additive in Stone Mastic Asphalt', International Journal of Engineering and Technology 2, no. 6, (2010): 379–387.
- [88] Mashaan, Nuha S., Ali Rezagholilou, Hamid Nikraz, and Amin Cheqenizadeh. "A review on using waste polymer as additive in asphalt mixture." International Journal of Advances in Science, Engineering and Technology (IJASEAT) 7, no. 2 (2019).
- [89] Mashaan, Nuha S., Ali Rezagholilou, and Hamid Nikraz. "Waste Plastic as Additive in Asphalt Pavement Reinforcement: A review." In AAPA International Flexible Pavements Conference, 18th, 2019, Sydney, New South Wales, Australia. (2019).
- [90] Mahrez, Abdelaziz, and Mohamed Rehan Karim. "Fatigue characteristics of stone mastic asphalt mix reinforced with fiber glass." International Journal of the Physical Sciences 5, no. 12 (2010): 1840-1847.
- [91] Masoudi, Sajjad, Sayyed Mahdi Abtahi, and Ahmad Goli. "Evaluation of electric arc furnace steel slag coarse aggregate in warm mix asphalt subjected to long-term aging." Construction and Building Materials 135 (2017): 260-266. doi:10.1016/j.conbuildmat.2016.12.177.
- [92] Ozden, Erdogan Baris, Ertug Erkus, Seref Agzikara, Erkan Evran, Ozgur Akdemir, and Ozgur Cobanoglu. "Yarns with conductive elastomeric cores, fabrics and garments formed of the same, and methods for producing the same." U.S. Patent Application 16/609,563, filed February 27, 2020.
- [93] S. F. Brown, R. D. Rowlett, and J. L. Boucher, 'Asphalt modification', in Highway research: sharing the benefits, Thomas Telford Publishing, 1991, pp. 181–203.

- [94] Mahrez, Abdelaziz, M. Karim, and H. Katman. "Prospect of using glass fiber reinforced bituminous mixes." Journal of the Eastern Asia Society for Transportation Studies 5 (2003): 794-807.
- [95] Munera, J. C., and E. A. Ossa. "Polymer modified bitumen: Optimisation and selection." Materials & Design (1980-2015) 62 (2014): 91-97. doi:10.1016/j.matdes.2014.05.009.
- [96] M. Park, J. Kwon, and S. Jeong, 'Display screen or portion thereof with animated graphical user interface'. Google Patents, 27-Oct-2015.
- [97] Otuoze, H.S., A.A. Shuaibu, H.A. Ahmed, I.T. Suleiman, I. Bello, and K.O. Yusuf. "Creep and Morphological Evaluation of Polypropylene Waste Modified Asphalt for Pavement Applications." Nigerian Journal of Technology 39, no. 3 (September 16, 2020): 654–664. doi:10.4314/njt.v39i3.3.
- [98] Yoo, Pyeong Jun, and Tae Woo Kim. "Strengthening of hot-mix asphalt mixtures reinforced by polypropylene-impregnated multifilament glass fibres and scraps." Construction and Building Materials 75 (2015): 415-420. doi:10.1016/j.conbuildmat.2014.11.009.
- [99] Huang, Xiao-Wei, Zhi-Qi Long, Liang-Shi Wang, and Zong-Yu Feng. "Technology development for rare earth cleaner hydrometallurgy in China." Rare Metals 34, no. 4 (2015): 215-222. doi:10.1007/s12598-015-0473-x.
- [100] Xue, Yongjie, Shaopeng Wu, Jun Cai, Min Zhou, and Jin Zha. "Effects of two biomass ashes on asphalt binder: Dynamic shear rheological characteristic analysis." Construction and Building Materials 56 (2014): 7-15. doi:10.1016/j.conbuildmat.2014.01.075.
- [101] Anurag, Kalia, Feipeng Xiao, and Serji N. Amirkhanian. "Laboratory investigation of indirect tensile strength using roofing polyester waste fibers in hot mix asphalt." Construction and building materials 23, no. 5 (2009): 2035-2040. doi:10.1016/j.conbuildmat.2008.08.018.
- [102] Chen, Xingwei, and Zhihong Xu. "Effects of load and temperature on permanent deformation properties of HMA." In Road Pavement Material Characterization and Rehabilitation: Selected Papers from the 2009 GeoHunan International Conference, pp. 49-55. 2009. doi:10.1061/41043(350)7.
- [103] Tapkin, Serkan. "The effect of polypropylene fibers on asphalt performance." Building and environment 43, no. 6 (2008): 1065-1071. doi:10.1016/j.buildenv.2007.02.011.
- [104] Elghatas, Hamza Mohamed Abdulhamid, Mohamad Yusri Bin Aman, and Ahmed Suliman B. Ali. "An Assessment on the Physical and Rheological Properties of Asphalt Binder Modified with Micro Bauxite Powder (MBP)." Lecture Notes in Civil Engineering (September 18, 2018): 30–37. doi:10.1007/978-981-13-2511-3\_4.
- [105] Zari, Maibritt Pedersen. "Biomimetic approaches to architectural design for increased sustainability." In The SB07 NZ Sustainable Building Conference, pp. 1-10. 2007.
- [106] Button, Joe W., and Robert L. Lytton. "Evaluation of fabrics, fibers and grids in overlays." In International Conference on the Structural Design. 1987.
- [107] You, Zhanping, Julian Mills-Beale, Justin M. Foley, Samit Roy, Gregory M. Odegard, Qingli Dai, and Shu Wei Goh. "Nanoclay-modified asphalt materials: Preparation and characterisation." Construction and Building Materials 25, no. 2 (2011): 1072-1078. doi:10.1016/j.conbuildmat.2010.06.070.
- [108] Jiang, Yi, and Rebecca S. McDaniel. "Application of cracking and seating and use of fibers to control reflective cracking." Transportation Research Record (1993): 150.
- [109] Guo, Qinglin, Lili Li, Yongchun Cheng, Yubo Jiao, and Chun Xu. "Laboratory evaluation on performance of diatomite and glass fiber compound modified asphalt mixture." Materials & Design (1980-2015) 66 (2015): 51-59. doi:10.1016/j.matdes.2014.10.033.
- [110] Zarei, Ali, Mohammad Zarei, and Omid Janmohammadi. "Evaluation of the effect of lignin and glass fiber on the technical properties of asphalt mixtures." Arabian journal for Science and engineering 44, no. 5 (2019): 4085-4094. doi:10.1007/s13369-018-3273-4.
- [111] Mashaan, Nuha Salim, Asim Hassan Ali, Mohamed Rehan Karim, and Mahrez Abdelaziz. "A review on using crumb rubber in reinforcement of asphalt pavement." The Scientific World Journal 2014 (2014). doi:10.1155/2014/214612.
- [112] Yao, Hui, Zhanping You, Liang Li, Xianming Shi, Shu Wei Goh, Julian Mills-Beale, and David Wingard. "Performance of asphalt binder blended with non-modified and polymer-modified nanoclay." Construction and Building Materials 35 (2012): 159-170. doi:10.1016/j.conbuildmat.2012.02.056.
- [113] Wang, Xi, Ruobing Wu, and Lulu Zhang. "Development and performance evaluation of epoxy asphalt concrete modified with glass fibre." Road Materials and Pavement Design 20, no. 3 (2019): 715-726. doi:10.1080/14680629.2017.1413006.

- [114] Fiore, V., T. Scalici, G. Di Bella, and A. Valenza. "A review on basalt fibre and its composites." Composites Part B: Engineering 74 (2015): 74-94. doi:10.1016/j.compositesb.2014.12.034.
- [115] Allen, R. Grover, Dallas N. Little, Amit Bhasin, and Charles J. Glover. "The effects of chemical composition on asphalt microstructure and their association to pavement performance." International Journal of Pavement Engineering 15, no. 1 (2014): 9-22. doi:10.1080/10298436.2013.836192.
- [116] Fang, Changqing, Xiaolong Liu, Ruien Yu, Pei Liu, and Wanqing Lei. "Preparation and properties of asphalt modified with a composite composed of waste package poly (vinyl chloride) and organic montmorillonite." Journal of Materials Science & Technology 30, no. 12 (2014): 1304-1310. doi:10.1016/j.jmst.2014.11.001.
- [117] Nuñez, Javier Yesid Mahecha, Matheus David Inocente Domingos, and Adalberto Leandro Faxina. "Susceptibility of lowdensity polyethylene and polyphosphoric acid-modified asphalt binders to rutting and fatigue cracking." Construction and Building Materials 73 (2014): 509-514. doi:10.1016/j.conbuildmat.2014.10.002.
- [118] Rahman, Md Nobinur, Mohammad Ahmeduzzaman, M. A. Sobhan, and T. U. Ahmed. "Performance evaluation of waste polyethylene and PVC on hot asphalt mixtures." American Journal of Civil Engineering and Architecture 1, no. 5 (2013): 97-102. doi:10.12691/ajcea-1-5-2.
- [119] Farrand, Bruce, and John Emery. "Recent improvements in quality of steel slag aggregate." Transportation research record 1468 (1995): 137-141.
- [120] Papayianni, Ioanna, Eleftherios Anastasiou, Michael Papachristoforou, and Alexandros Liapis. "Steel Slag Concrete for Pavement Construction." (2016).
- [121] A El-Assaly, Ashraf, and Ralph Ellis. "Evaluation of recycling waste materials and by-products in highway construction." The International Journal of Sustainable Development & World Ecology 8, no. 4 (2001): 299-308. doi:10.1080/13504500109470088.
- [122] Prezzi, Monica, Paola Bandini, J. Antonio H. Carraro, and Paulo JM Monteiro. "Use of recyclable materials in sustainable Civil engineering applications." (2011). doi:10.1155/2011/896016.
- [123] Vazquez-Calvo, C., M. Alvarez De Buergo, and R. Fort. "Overview of recent knowledge of patinas on stone monuments: the Spanish experience." Geological Society, London, Special Publications 271, no. 1 (2007): 295-307. doi:10.1144/gsl.sp.2007.271.01.28.
- [124] Motz, Holger, and Jens Geiseler. "Products of steel slags an opportunity to save natural resources." Waste management 21, no. 3 (2001): 285-293. doi:10.1016/s0956-053x(00)00102-1.
- [125] Hainin, Mohd Rosli, Nur Izzi Md Yusoff, Mohd Fahmi Mohammad Sabri, Mohd Azizi Abdul Aziz, Mohd Anwar Sahul Hameed, and Wasid Farooq Reshi. "Steel slag as an aggregate replacement in Malaysian hot mix asphalt." ISRN Civil Engineering 2012 (2012). doi:10.5402/2012/459016.
- [126] Xue, Yongjie, Shaopeng Wu, Haobo Hou, and Jin Zha. "Experimental investigation of basic oxygen furnace slag used as aggregate in asphalt mixture." Journal of hazardous materials 138, no. 2 (2006): 261-268. doi:10.1016/j.jhazmat.2006.02.073.
- [127] Abdullah, Mohd Ezree, Kemas Ahmad Zamhari, Mohd Rosli Hainin, Ebenezer Akin Oluwasola, Norhidayah Abdul Hassan, and Nur Izzi Md Yusoff. "Engineering properties of asphalt binders containing nanoclay and chemical warm-mix asphalt additives." Construction and Building Materials 112 (2016): 232-240. doi:10.1016/j.conbuildmat.2016.02.089.
- [128] G. H. Shafabakhsh and O. J. Ani, 'Experimental investigation of effect of Nano TiO2/SiO2 modified bitumen on the rutting and fatigue performance of asphalt mixtures containing steel slag aggregates', Construction and Building Materials, vol. 98, pp. (2015): 692–702. doi:10.1016/j.conbuildmat.2015.08.083.
- [129] Yildirim, Irem Zeynep, and Monica Prezzi. "Chemical, Mineralogical, and Morphological Properties of Steel Slag." Advances in Civil Engineering 2011 (2011): 1–13. doi:10.1155/2011/463638.
- [130] Liu, Wenhuan, Hui Li, Huimei Zhu, and Pinjing Xu. "Properties of a steel slag-permeable asphalt mixture and the reaction of the steel slag-asphalt interface." Materials 12, no. 21 (2019): 3603. doi:10.3390/ma12213603.
- [131] Muniandy, R, E Aburkaba, and R Taha. "Effect of Mineral Filler Type and Particle Size on the Engineering Properties of Stone Mastic Asphalt Pavements." The Journal of Engineering Research [TJER] 10, no. 2 (December 1, 2013): 13. doi:10.24200/tjer.vol10iss2pp13-32.
- [132] Noureldin, A. Samy, and Rebecca S. McDaniel. "Evaluation of surface mixtures of steel slag and asphalt." Transportation Research Record 1269 (1990).
- [133] Modarres, Amir, and Morteza Rahmanzadeh. "Application of coal waste powder as filler in hot mix asphalt." Construction and Building Materials 66 (2014): 476-483. doi:10.1016/j.conbuildmat.2014.06.002.

- [134] Tapkin, Serkan. "Mechanical Evaluation of Asphalt-aggregate Mixtures Prepared with Fly Ash as a Filler Replacement." Canadian Journal of Civil Engineering 35, no. 1 (January 2008): 27–40. doi:10.1139/107-082.
- [135] Santos, João, Gerardo Flintsch, and Adelino Ferreira. "Environmental and Economic Assessment of Pavement Construction and Management Practices for Enhancing Pavement Sustainability." Resources, Conservation and Recycling 116 (January 2017): 15–31. doi:10.1016/j.resconrec.2016.08.025.
- [136] Pouranian, M. Reza, and Mehdi Shishehbor. "Sustainability assessment of green asphalt mixtures: A review." Environments 6, no. 6 (2019): 73. doi:10.3390/environments6060073.
- [137] Maghool, F., Arulrajah, A., Du, Y. J., Horpibulsuk, S., & Chinkulkijniwat, A. (2017). Environmental impacts of utilising waste steel slag aggregates as recycled road construction materials. Clean Technologies and Environmental Policy, 19(4), 949-958. doi:10.1007/s10098-016-1289-6.